

Citation: Verrest L, Monnerat S, Musa AM, Mbui J, Khalil EAG, Olobo J, et al. (2024) *Leishmania* blood parasite dynamics during and after treatment of visceral leishmaniasis in Eastern Africa: A pharmacokinetic-pharmacodynamic model. PLoS Negl Trop Dis 18(4): e0012078. https://doi.org/ 10.1371/journal.pntd.0012078

Editor: Helen P. Price, Keele University, UNITED KINGDOM

Received: May 9, 2023

Accepted: March 18, 2024

Published: April 19, 2024

Copyright: © 2024 Verrest et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be made publicly available as the dataset contains sensitive and personal information, which cannot be anonymized due to the nature of the data. Requests to access the data may be sent to dataoffice@uu. se.

Funding: This work was supported through DNDi by the European Union Seventh Framework Programme Africoleish (grant number 305178); the European Union – Framework Programme 7 RESEARCH ARTICLE

Leishmania blood parasite dynamics during and after treatment of visceral leishmaniasis in Eastern Africa: A pharmacokineticpharmacodynamic model

Luka Verrest¹, Séverine Monnerat², Ahmed M. Musa³, Jane Mbui⁴, Eltahir A. G. Khalil³, Joseph Olobo⁵, Monique Wasunna⁶, Wan-Yu Chu^{1,7}, Alwin D. R. Huitema^{1,8,9}, Henk D. F. H. Schallig¹⁰, Fabiana Alves², Thomas P. C. Dorlo^{1,7}*

 Department of Pharmacy & Pharmacology, Antoni van Leeuwenhoek Hospital/Netherlands Cancer Institute, Amsterdam, the Netherlands, 2 Drugs for Neglected Diseases initiative, Geneva, Switzerland,
Institute of Endemic Diseases, University of Khartoum, Khartoum, Sudan, 4 Centre for Clinical Research, Kenya Medical Research Institute, Nairobi, Kenya, 5 Department of Medical Microbiology, College of Health Sciences, Makerere University, Kampala, Uganda, 6 Drugs for Neglected Diseases initiative (DNDi), Nairobi, Kenya, 7 Department of Pharmacy, Uppsala University, Uppsala, Sweden, 8 Department of Clinical Pharmacy, University Medical Center Utrecht, Utrecht University, Utrecht, The Netherlands, 9 Department of Pharmacology, Princess Máxima Center for Pediatric Oncology, Utrecht, The Netherlands, 10 Department of Medical Microbiology and Infection Prevention, Laboratory for Experimental Parasitology, Academic Medical Center, Amsterdam, the Netherlands

* thomas.dorlo@farmaci.uu.se

Abstract

Background

With the current treatment options for visceral leishmaniasis (VL), recrudescence of the parasite is seen in a proportion of patients. Understanding parasite dynamics is crucial to improving treatment efficacy and predicting patient relapse in cases of VL. This study aimed to characterize the kinetics of circulating *Leishmania* parasites in the blood, during and after different antileishmanial therapies, and to find predictors for clinical relapse of disease.

Methods

Data from three clinical trials, in which Eastern African VL patients received various antileishmanial regimens, were combined in this study. *Leishmania* kinetoplast DNA was quantified in whole blood with real-time quantitative PCR (qPCR) before, during, and up to six months after treatment. An integrated population pharmacokinetic-pharmacodynamic model was developed using non-linear mixed effects modelling.

Results

Parasite proliferation was best described by an exponential growth model, with an *in vivo* parasite doubling time of 7.8 days (RSE 12%). Parasite killing by fexinidazole, liposomal amphotericin B, sodium stibogluconate, and miltefosine was best described by linear models directly relating drug concentrations to the parasite elimination rate. After treatment,

(grant agreement 305178); the World Health Organization - Special Programme for Research and Training in Tropical Diseases (WHO-TDR); the French Development Agency (AFD), France (grant number CZZ2062); UK aid, UK; the Federal Ministry of Education and Research (BMBF) through KfW, Germany; the Medicor Foundation; Médecins Sans Frontières International; the Swiss Agency for Development and Cooperation, Switzerland (grant number 81017718 and 642.33/2010/0779/02 NID); the Dutch Ministry of Foreign Affairs, the Netherlands (grant number PDP15CH21); the Spanish Agency for International Development Cooperation (AECID) and other private foundations and individuals. T.P.C.D. is personally supported by the ZonMw/Dutch Research Council (NWO) Veni grant (project no. 91617140) and the Swedish Research Council (VR 2022-01251). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

parasite growth was assumed to be suppressed by the host immune system, described by an E_{max} model driven by the time after treatment. No predictors for the high variability in onset and magnitude of the immune response could be identified. Model-based individual predictions of blood parasite load on Day 28 and Day 56 after start of treatment were predictive for clinical relapse of disease.

Conclusion

This semi-mechanistic pharmacokinetic-pharmacodynamic model adequately captured the blood parasite dynamics during and after treatment, and revealed that high blood parasite loads on Day 28 and Day 56 after start of treatment are an early indication for VL relapse, which could be a useful biomarker to assess treatment efficacy of a treatment regimen in a clinical trial setting.

Author summary

After primary treatment of visceral leishmaniasis (VL), relapse of disease occurs in a proportion of patients. Understanding parasite dynamics is crucial to improving treatment efficacy and predicting disease relapse at an early stage. *Leishmania* kinetoplast DNA loads in blood from East African VL patients from three clinical trials, treated with five different treatment regimens, were used to develop a semi-mechanistic model to integrate *in vivo* parasite replication in the host, parasite clearance by different VL drug regimens, and suppression of parasite regrowth by the host immune system after treatment. This model deepened our understanding of the parasite-drug-host interaction and described the *in vivo* parasite growth rate in human for the first time. Moreover, the model revealed that high blood parasite loads on Day 28 and Day 56 after start of treatment are an early indication for VL relapse, which could serve as a biomarker to predict long-term clinical outcome, an important new tool for optimizing future VL treatment regimens.

1. Introduction

In Eastern Africa, the region with the highest visceral leishmaniasis (VL) incidence globally, efficacy rates of currently available VL therapies range from 72% to 91% [1–3]. Almost all patients show a good initial response to drug treatment with improvement in the clinical signs and symptoms and a negative parasitological test of cure by microscopy at the end of treatment, fulfilling the definition of initial cure. Therapy failure occurs mainly by relapse of disease after initial successful treatment due to parasite recrudescence, which is a long-term event that can occur up to 12 months after treatment, or even longer [4]. Following successful treatment of VL infection, latent parasites may still be present and can be reactivated, resulting in recurrence of VL once immunity is compromised [5–8]. Patients who start VL treatment are mostly malnourished and severely sick, often presenting fever, hypergammaglobulinaemia, and haematological depletions such as anaemia, neutropaenia, and leucopaenia [9]. These complications may lead to an impaired functioning of the immune system, which is unable to suppress or eradicate *Leishmania* parasites [10]. Once the patient improves after the start of treatment, the immune system can recover and clear or control the parasites.

It is particularly difficult to predict which patients will relapse, as almost all patients are clinically cured at the end of treatment. Quantification of the total *Leishmania* parasite burden

in patients might give a good approximation of the severity of disease and response to treatment. The gold standard in clinical trials is quantification of parasites by microscopy in aspirates from spleen or bone marrow, where the mainstay of parasites is resident, performed before start of treatment to confirm VL infection, at the end of treatment to assess initial cure, and when relapse is suspected. However, aspiration is a highly invasive procedure and is, therefore, not suitable for more regulator monitoring. Quantification of circulating Leishmania kinetoplast DNA (kDNA) in blood by real-time quantitative PCR (qPCR) is an attractive patient-friendly alternative, which allows the collection of longitudinal data. Previously, qPCR parasite load in blood showed a good correlation with qPCR parasite load in aspirated organ tissue ($\rho = 0.80$) [11], indicating that whole blood is an adequate proxy compartment for monitoring parasite biomass in the infected organs. Positive blood parasite load after treatment has been associated with a higher risk of VL relapse [12-24], and the blood parasite load on day 56 after start of treatment has been found to be a highly sensitive predictor of relapse at a cut-off of 20 parasites/mL [11]. This low cut-off value indicates that very low blood parasite loads are already associated with a higher risk of disease relapse, even when patients do not yet present reoccurrence of clinical symptoms.

The phenomenon of asymptomatic parasite recrudescence without clinical relapse complicates the analysis of parasite dynamics in relation to clinical response [11]. Analysis of parasite dynamics is further complicated by the large baseline variability of parasite load and the large inter-patient variability in response [11]. Factors that affect parasite dynamics and treatment response may depend on the initial degree of parasite depletion by the treatment, but also on parasite-related factors such as of the total burden of parasites present at the start of treatment or the virulence of the parasite. Sufficient suppression of the parasite by a properly functioning host immune system is of crucial importance for achieving long-lasting cure; this might be affected by patient-specific factors such as severity of VL infection, co-infections such as HIV, or malnutrition.

A dynamic pharmacokinetic-pharmacodynamic model is needed to capture the interplay between parasite growth, drug exposure, drug-induced parasite clearance, and suppression of parasite regrowth after treatment of the host. While pharmacokinetic-pharmacodynamic models have been shown to be useful for the understanding of other parasitic diseases, such as malaria [25–28], *in vivo* parasite replication rates or parasite clearance by VL drugs have not previously been studied and quantified for *Leishmania*. Longitudinal analysis of repeated blood parasite loads during and after VL treatment will enable characterization of these dynamics [11].

In this study, we aimed to develop a pharmacokinetic-pharmacodynamic model of *Leish-mania* blood parasite loads in VL patients receiving various drug regimens, to get a better understanding of the interplay between parasite, drugs, and host. Moreover, using a semi-mechanistic model we characterized the different effects of VL treatment on parasite clearance during treatment, to further enable optimization of dosing regimens or new combination regimens. Lastly, to predict the long-term response of *Leishmania* parasites to treatment, we aimed to identify early biomarkers or model-derived predictors for parasite recrudescence and clinical relapse of disease.

2. Methods

2.1 Ethics statement

The samples analysed originated from three Phase II open-label randomized clinical trials, that assessed the safety and efficacy of different VL treatment regimens in Eastern Africa (Table 1). Ethical approval was obtained from independent ethics committee at the Faculty of Medicine,

Study	LEAP0208			LEAP0714	FEXI-VL-001	Total
Treatment	AmB+SSG10D	AmB+MF10D	MFC28D	MFA28D	Fexi10D	
Patients ^a (n)	40	44	46	29	13	172
Cured patients (<i>n</i> [%])	37 (93)	37 (84)	35 (76)	27 (93)	2 (15)	138 (80)
Male (<i>n</i> [%])	28 (70)	36 (82)	41 (89)	8 (28)	1 (8)	114 (66)
Age (years) ^b	14 (7-40)	14 (7-30)	15 (7–37)	8 (4–12)	26 (16-50)	14 (4–50)
Body weight (kg) ^b	33 (15-69)	34 (15–57)	36 (16-60)	22 (13-30)	51 (42-64)	34 (13-69)
PD samples ^c collected (n)	306	306	313	208	127	1260
PD samples ^c included (n [% excluded])	243 (21)	256 (16)	278 (11)	138 (34)	77 (39)	992 (21)

Table 1. Patient characteristics and data overview.

AmB+MF10D: Amphotericin B 10 mg/kg (1 day) + miltefosine 2.5 mg/kg (10 days); AmB+SSG10D: Amphotericin B 10 mg/kg (1 day) + SSG 20 mg/kg (10 days); Fexi10D: Fexinidazole 1800 mg (4 days), 1200 mg (6 days); MFC28D: Miltefosine conventional 2.5 mg/kg (28 days), MFA28D: Miltefosine allometric dosing (28 days) ^a Patients included in the analysis

^b Mean (range) at baseline

^c Samples for quantification of blood parasite load by qPCR

https://doi.org/10.1371/journal.pntd.0012078.t001

University of Khartoum and the Sudanese National Medicines and Poisons Board in Sudan, the London School of Hygiene & Tropical Medicine Ethics Committee (#5543 and #6351), a 'declaration of no objection' from the Academic Medical Center Medical Ethics Committee (LEAP0208), and from institutional ethics committees at the Kenya Medical Research Institute and at Makerere University, Uganda (LEAP0208, LEAP0714), and the Institute of Endemic Diseases at the University of Khartoum (Fexi-VL-001). Study participants or their parents/ guardians (for children under 18 years) provided written informed consent before enrolment into the study, including participation in the pharmacokinetic and parasitological assessments.

2.2 Study design, patients, and clinical assessment of efficacy

In clinical trial LEAP0208 (NCT01067443 [1]) three different treatment regimens in patients from Kenya (Kimalel) and Sudan (Dooka and Kassab) were compared: (1) a combination of 10 mg/kg liposomal amphotericin B (IV) on day 1 followed by 10 days of 20 mg/kg sodium stibogluconate (IM) from day 2-11 (n = 51) (AmB+SSG10D), (2) a combination of 10 mg/kg liposomal amphotericin B (AmBisome) (IV) on day 1 followed by 10 days of 2.5 mg/kg/day (maximum 150 mg/day) miltefosine (oral) from day 2-11 (n = 49) (AmB+MF10D), (3) monotherapy of conventional miltefosine dosing for 28 days of 2.5 mg/kg/day (maximum 150 mg/ day) (n = 51) (MFC28D). In clinical trial LEAP0714, (NCT02431143 [2]) 30 paediatric patients from Kenya (Kacheliba) and Uganda (Amudat) who received allometric dosing of miltefosine monotherapy for 28 days (ranging between 30 and 100 mg/day) were studied (MFA28D). In clinical trial FEXI-VL-001 (NCT01980199) a flat dosing of 1800 mg/day fexinidazole for 4 days, followed by 1200 mg/day for 6 days was investigated in 14 adult patients from Sudan (Dooka) (Fexi10D). Patients included in all studies showed VL clinical symptoms (fever and splenomegaly) and had a confirmatory parasitological microscopic diagnosis. All recruited patients were HIV negative. None presented with severe VL, suffered severe malnutrition, or any serious underlying disease or concomitant severe infection at the time of diagnosis.

Clinical assessment of efficacy was defined by initial cure: a negative parasitological test of cure by microscopy on Day 28; final cure: initial cure and absence of VL signs and symptoms until Day 210 (6 months), *i.e.*, no occurrence of relapse during the follow-up period and no requirement for rescue treatment; or initial treatment failure: patients who died or required rescue treatment before completion of study treatment.

2.3 Sample collection, bioanalysis, and data exclusion

Pharmacokinetic samples were collected for miltefosine (LEAP0208 and LEAP0714) and fexinidazole and its active metabolites fexinidazole sulfoxide (M1) and fexinidazole sulfone (M2) (FEXI-VL-001). Miltefosine sample collection was performed in all patients receiving miltefosine monotherapy or a combination therapy with miltefosine. Sample collection and bioanalysis has been described previously [1,29,30]. In FEXI-VL-001, dried-blood spot (DBS) samples were collected at multiple time points during treatment in all patients receiving fexinidazole, with more dense sampling on Day 1 and Day 10 of treatment. Whole blood concentrations of fexinidazole, M1, and M2 were quantified, as described in published literature [31]. Whole blood EDTA samples with a volume of 200 µL were collected for pharmacodynamic assessment in all patients prior to treatment and on Days 3, 7, 14, 28, 56, and 210 (LEAP0208), on Days 3, 7, 14, 21, 28, and 56 (LEAP0714), and on Days 1, 3, 5, 8, 11, 14, 28, 56, and 210 (FEX-I-VL-001). Leishmania kinetoplastid DNA (kDNA) was quantified in these samples using a qPCR method, to determine the parasite load in the patient (hereafter referred to as blood parasite load). Human glyceraldehyde 3-phosphate dehydrogenase (GAPDH) DNA was used as an internal control for validation of the extraction procedure. The lower limit of quantification (LLOQ) was set to 1 parasite/mL. A detailed description of the DNA extraction, primers used, and qPCR protocol has been described earlier [11].

qPCR data were excluded from the analysis for patients who were considered initial treatment failures and did not complete the study treatment, and for samples collected after rescue treatment was given to relapsed patients, and for samples considered to be unreliable (*i.e.*, if the GAPDH response indicated unreliable or incomplete DNA extraction of the sample, if the sample quality was low, or if there was insufficient sample material). In addition, qPCR data were excluded if values were physiologically implausible, *i.e.* if baseline parasite loads were exceedingly low (<100 parasites/mL), while later parasite loads in the same individual in the first week of treatment indicated much higher loads. Furthermore, patients who received miltefosine or fexinidazole for whom no pharmacokinetic data were available were also excluded.

2.4 Population pharmacokinetic-pharmacodynamic analysis

An integrated pharmacokinetic-pharmacodynamic model was developed using the nonlinear mixed effects modelling software NONMEM (version 7.5, ICON Development Solutions, USA) using the first-order conditional estimation method with interaction (FOCE-I). Data cleaning and visualization of the data were performed using R (version 4.0.2), and model evaluation was done using Perl-speaks-NONMEM (PsN, version 5.2.6) and the interface Pirana (version 3.0.0). Model development was performed in three consecutive steps: 1) characterization of parasite growth, 2) characterization of drug-induced parasite clearance during treatment, and 3) characterization of host-induced parasite suppression after treatment. In each step, a relevant subset of the data was used to estimate the different model parameters. Blood parasite loads below the limit of quantification (BLQ) were fixed to half the BLQ (0.5 parasite/mL). Between-subject variability (BSV) was evaluated for all parameters and implemented using an exponential error model. Residual variability was modeled using combined additive and proportional error models, with the additive error fixed to BLQ/2 (0.5 parasite/mL).

2.4.1 Parasite growth. No pharmacodynamic data was available before start of treatment, complicating characterization of natural parasite growth in primary VL patients. *In vivo* parasite growth was, therefore, modelled and estimated based on data from patients treated with fexinidazole, where an inadequate drug response that resulted in rapid parasite recrudescence directly after treatment was observed in the majority of patients. To describe parasite proliferation, turn-over and exponential growth models were evaluated.

2.4.2 Drug-induced parasite clearance during treatment. Previously developed population pharmacokinetic models of miltefosine [29] and fexinidazole and its active metabolites M1 and M2 (internal report) were used to derive individual predicted pharmacokinetic concentrations, which were used as pharmacokinetic input to the model. For liposomal amphotericin B and SSG, a kinetic-pharmacodynamic approach was used assuming a one compartment model with first-order elimination, using the administered drug amounts and previously reported elimination half-lives of 6 hours for liposomal amphotericin B [32] and 2 hours for SSG [33,34]. Direct and delayed sigmoidal E_{max} and linear models were evaluated to model drug-induced killing of parasites driven by individual predicted concentration-time curves for each drug.

2.4.3 Host-induced parasite suppression after treatment. Visual inspection of individual blood parasite loads over time indicated diverse and highly variable profiles after treatment (Fig A in <u>S1 File</u>). Typical profiles after treatment included 1) complete parasite clearance with no parasite recrudescence during follow-up (complete parasitological cure) 2) initial parasite clearance followed by parasite recrudescence, where parasite regrowth is initiated at different time points during the follow-up period, and 3) initial parasite clearance followed by parasite recrudescence early after treatment, followed by parasite clearance later during follow-up (Table A in <u>S1 File</u>).

The aim of this part of building the model was to enable description of these three distinct typical parasite load-time profiles during the follow-up phase after treatment. Based on the hypothesis that parasite suppression after treatment, in the absence of drug pressure, is driven by the host immune system, the variable onset in parasite suppression and clearance after the end of treatment by the host immune system was implemented by a sigmoidal E_{max} function, driven empirically by the time after treatment, which captured differences in onset and magnitude of parasite recrudescence among patients (Eq 1). To allow for clinically meaningful parasite recrudescence after complete drug-induced parasite depletion, the parasite compartment was restricted to ≥ 1 parasite/mL.

$$k_{imm} = \frac{I_{max} * Time^{\gamma}}{IT_{50}^{\gamma} + Time^{\gamma}}$$
 Eq 1

In Eq 1, k_{imm} is parasite suppression by the immune system, Time is the time after start of treatment, I_{max} is the maximum inhibitory effect of the immune system, and IT_{50} is the time after treatment of half the maximal immune effect. The parameter γ denotes the steepness of the time-effect relationship.

With the aim of finding biomarkers predictive of (suppression of) parasite recrudescence after therapy, haematological and biochemical markers were evaluated as drivers of parasite suppression (k_{imm}) rather than time after treatment. Available haematological and biochemical data included haemoglobin, white blood cells, platelets, and creatinine for all patients, neutrophils, lymphocytes, and monocytes (LEAP0714 and FEXI-VL-001), and total protein and albumin (FEXI-VL-001). All these covariates were evaluated graphically to assess their relationship with parasite recrudescence and suppression. White blood cell count was a physiologically plausible covariate as it may reflect the overall activity of the immune system. Therefore, white blood cell counts on Day 56, either absolute or relative to baseline, was evaluated on parasite suppression by two strategies. First, white blood cell count was evaluated as covariate on the extent of suppression of parasite regrowth after treatment (Eq 1, I_{max}) and the onset of parasite suppression after treatment (Eq 1, $I_{T_{50}}$) using a slope function (Eq 2).

$$P_{TV} = P_{pop} * (1 + (WBC_i - WBC_{med}) * l)$$
 Eq 2

where P_{TV} is the typical parameter value for IT_{50} or I_{max} at covariate value WBC_i, P_{pop} the population estimate of IT_{50} or I_{max} , WBC_i the white blood cell count or the white blood cell count relative to baseline for individual i on Day 56, WBC_{med} the median covariate value in the population, and *l* the slope factor.

Second, white blood cell count was used as a descriptor for assessing parasite suppression, instead of using empirical time after treatment (Eq.3).

$$k_{imm} = \frac{I_{max} * WBC_i^{\gamma}}{IC_{50}^{\gamma} + WBC_i^{\gamma}}$$
 Eq 3

In Eq.3, k_{imm} is parasite suppression by the immune system, WBC_i the white blood cell count or the white blood cell count relative to baseline for individual i on Day 56, I_{max} is the maximum inhibitory effect of the immune system, and IC₅₀ is the white blood cell count of half the maximal immune effect. The parameter γ denotes the steepness of the time-effect relationship.

2.5 Model evaluation

The aim of this modeling analysis was to describe parasite clearance rates for the different treatment regimens, as well as to capture the different parasite profiles after treatment, which are driven by the immune system. Standard goodness of fit plots and individual prediction plots were used to guide model development. Model selection was based on scientific plausibility and minimum objective function value (OFV), where a decrease of 6.63 points in OFV corresponding to a P value <0.01 was considered significant, with 1 degree of freedom following a χ^2 distribution. Precision in parameter estimates was obtained by sampling importance resampling (SIR). To further assess the goodness of fit and to compare the model description of parasite clearance during the treatment period in more detail, a visual predictive check of blood parasite loads during treatment was performed for all treatment regimens, in which the effects of both drugs' PK, parasite growth and the immune response are included. Simulations of parasite profiles were performed in a typical patient receiving different treatment regimens, and having different immune system activities, to illustrate the effect of different drugs and different immune effects.

2.6 Evaluation of the correlation between parasite exposure and clinical outcome

One of the goals of the model was to evaluate the predictiveness of parasite load or parasite clearance for subsequent clinical relapse of VL during follow-up. To characterize the relationship of early drug-induced parasite clearance and long-term clinical outcome, different parasite exposure metrics during and early after treatment were evaluated. The sensitivity and specificity of predicting clinical relapse based on blood parasite load on Day 10, Day 28, and Day 56 and the optimal cutoff values were derived by receiver-operating characteristic (ROC) curves, generated using the R packages "pROC" and "plotROC". The area under the ROC curves (AUC_{ROC}) were compared to find the most predictive parameter for clinical relapse in terms of follow-up day (Day 10, Day 28, or Day 56). The fraction of patients having a blood parasite load below the optimal cutoff value resulting from the ROC curve was compared between treatment regimens and between cured and relapsed patients. Second, model-based predictions of parasite load and parasite exposure during and early after treatment (at baseline and Day 10, 28, and 56) were compared between treatment regimens and to clinical outcome. Cumulative exposure to parasites during and after treatment, expressed as the area under the blood parasite load-time curve (AUC), was also compared as a measure of parasite clearance during therapy. This parameter incorporates both the drug-induced parasite clearance rate and the baseline parasite load, which was highly variable among patients. The parasite AUC during the first 10, 28, and 56 days were compared. I_{max} and IT_{50} were compared to assess any treatment-related effects on the suppression of parasite recrudescence after treatment.

Results

3.1 Data

In total, 1260 blood samples from 188 patients were available for qPCR analysis (Table 1). The following qPCR observations were excluded from the analysis: 2 initial treatment failures (7 observations), 21 observations collected after rescue treatment, 190 unreliably extracted samples, and 44 physiologically impossible observations (3.5%). One patient who received AmB +MF10D but lacking pharmacokinetic data was excluded (6 observations). Of the 992 blood parasite load observations included in the analysis, 359 observations (36%) were BLQ. An overview of the data included in the analysis up to Day 56 is shown in Fig 1, which displays trends in the data during and early after treatment between the different regimens.

3.2 Population pharmacokinetic-pharmacodynamic model

3.2.1 Parasite growth and drug-dependent parasite clearance. Parasite proliferation was best described by an exponential growth model (Fig 2), with an *in vivo* parasite doubling time of 7.8 days (RSE 12%) (Table 2 and Fig 2). The model-derived individual predictions for the Fexi10D regimen indicated that parasite proliferation was adequately described, as the model captured parasite growth after treatment in these patients (Fig A in S1 File). Drug-dependent parasite killing was best described by first-order linear pharmacokinetic-pharmacodynamic models (fexinidazole and miltefosine) or kinetic-pharmacodynamic models (liposomal amphotericin B and SSG), where the parasite killing rate was directly proportional to the drug concentration. The visual predictive check shows that the model quite adequately described parasite clearance in the majority of treatment regimens (Fig 3). There were some discrepancies between the observed data and the model simulations, i.e., parasite recrudescence in Fexi10D was under-predicted by the model, baseline blood parasite load in MFA28D was under-predicted, and the parasite clearance rate in AmB+MF10D was under-predicted,



Fig 1. Blood parasite loads in Eastern African visceral leishmaniasis patients during treatment and early follow-up. Depicted are median (IQR) observed blood parasite loads coloured by treatment outcome (cured patients in red and relapsed patients in blue), stratified by treatment regimen. AmB+SSG10D: 10 mg/kg amphotericin B (day 1) + 20 mg/kg/day SSG (day 2 to 11); AmB+MF10D: 10 mg/kg amphotericin B (day 1) + 100 mg/day miltefosine (day 2 to 11); MFC28D: miltefosine conventional dose (100 mg/day) (28 days); MFA28D: miltefosine allometric dose (28 days); Fexi10D: 1800 mg/day fexinidazole (4 days) + 1200 mg/day fexinidazole (6 days). Gray dashed lines represent the end of treatment.

https://doi.org/10.1371/journal.pntd.0012078.g001



Fig 2. Schematic overview of the final pharmacokinetic-pharmacodynamic model, exemplified with the pharmacokinetic model for fexinidazole and its active metabolites M1 and M2. In the parasite model, k_{GR} is the parasite replication rate, k_{drug} is the drug-driven parasite clearance rate, λ_{drug} the drug-specific linear effect, and C_{drug} the drug concentration of either miltefosine, the sum of M1 and M2 for fexinidazole, amphotericin B, or SSG. k_{IMM} is the immune-driven parasite clearance, I_{max} the maximum inhibition by the immune response, IT_{50} the time at half-maximum inhibition, and γ the steepness of the time-effect relationship, which was empirically fixed to 5. In the fexinidazole pharmacokinetic model, CL is the clearance of fexinidazole, M1 or M2, and V₂, V₃, and V₄ the volume of distribution of fexinidazole, M1, and M2, respectively.

especially the upper boundary of the prediction interval. The time course of drug exposure in the different treatment regimens was simulated for a typical patient (Fig 4), illustrating variable durations of exposure depending on the pharmacokinetic characteristics of the drugs and their corresponding elimination half-lives, which were much longer for miltefosine. This resulted in

Parameter	Parameter estimate	RSE (%) ^a	BSV (CV%)	RSE (%) ^a	Shrinkage (%)
E _{BASE} (parasites/mL)	5324	16	243	12	3
k_{GR} (h ⁻¹)	0.0037	12	-	-	-
$\lambda_{\text{fexi}} (\mu g^{-1*} L^* h^{-1})$	0.0011	15	44	59	3
$\lambda_{\rm MF} (\mu g^{-1*} L^* h^{-1})$	0.0010	5	42	21	5
$\lambda_{\rm Amb} (mg^{-1*}kg^*h^{-1})$	0.0245	8	-	-	-
$\overline{\lambda_{SSG} (mg^{-1*}kg^*h^{-1})}$	0.0112	5	-	-	-
I_{max} (h ⁻¹)	0.037 (fixed) ^b	-	298	16	52
IT ₅₀ (days)	33.7	0.1	230	12	40
Proportional residual error (%)	101	-		0.7	18
Additive residual error (parasites/mL)	0.5 (fixed) ^c	-		-	18

Table 2. Parameter estimates for the final population pharmacokinetic-pharmacodynamic model.

BSV: between-subject variability; CV: coefficient of variation; E_{BASE} : Baseline blood parasite load; I_{max} : maximum inhibition by immune response; IT_{50} : time at half-maximum inhibition; k_{GR} : parasite growth constant; λ_{Amb} : linear drug effect Amphotericin B; λ_{fexi} : linear drug effect fexinidazole; λ_{MF} : linear drug effect miltefosine; λ_{SSG} : linear drug effect SSG; RSE: relative standard error

^a Obtained by SIR

 $^{\rm b}$ Fixed to ten times $k_{\rm GR}$

^c Fixed to half the lower limit of quantification

https://doi.org/10.1371/journal.pntd.0012078.t002



Fig 3. Prediction-corrected visual predictive checks for the final pharmacokinetic-pharmacodynamic blood *Leishmania* parasite load model until Day 56 after start of treatment. Solid lines: median of the observed values; dashed lines: the 10th and 90th percentiles of the observed values; dark and light blue areas: the 90% confidence intervals of the simulated median and percentiles, based on 1000 simulations.

https://doi.org/10.1371/journal.pntd.0012078.g003



Fig 4. Simulations of typical pharmacokinetic profiles of patients receiving A) 1800 mg/day fexinidazole (4 days) + 1200 mg/day fexinidazole; B) 10 mg/kg amphotericin B (day 1) + 20 mg/kg/day SSG (day 2 to 11); C) 10 mg/kg amphotericin B (day 1) + 100 mg/day miltefosine (day 2 to 11); or D) 100 mg/day miltefosine (28 days).

persistent drug exposure of about 11 days for the AmB+SSG10D and Fexi10D regimens, 30 days for the AmB+MF10D regimen, and 50 days for the MFC28D and MFA28D regimens. The individual model-based predictions during treatment further illustrate different parasite clearance rates for the various drug regimens (Fig A and Fig B in S1 File). The differential pharmacokinetic profiles, in combination with the drug-specific drug effects on parasite clearance, led to different patterns of parasite dynamics during the treatment period for each treatment regimen (Fig 5A). Rapid and effective parasite clearance was induced by treatment with liposomal amphotericin B (AmB+SSG10D and AmB+MF10D), while a slow onset with later parasite clearance was observed for miltefosine (MFC28D and MFA28D), and a weak response was observed for fexindazole (Fexi10D).

3.2.2 Parasite suppression after treatment. Parasite suppression after treatment was empirically described by a first-order elimination process with a sigmoid E_{max} function driven by the time after treatment, with γ empirically fixed to 5, representing a fast onset of parasite



Fig 5. A (left plot): Simulation of drug effects of different VL therapies of typical patients receiving 1) 10 mg/kg amphotericin B (day 1) + 20 mg/kg/day SSG (day 2 to 11) (blue curve), 2) 10 mg/kg amphotericin B (day 1) + 100 mg/day miltefosine (day 2 to 11) (grey curve), 3) 100 mg/day miltefosine (28 days) (red curve), or 4) 1800 mg/day fexinidazole (4 days) + 1200 mg/day fexinidazole (6 days) (yellow curve). No immune response after the end of treatment was observed. Other parameters were fixed to the population values. B (right plot): Simulation of typical patients receiving 150 mg/day miltefosine for 28 days. Patients have an IT_{50} of 1000 h (blue curve), 5000 h (grey curve), and 100,000 h (red curve). Other parameters were fixed to the population values.

suppression by the host's immune system after start of treatment (Fig 2). The model captured the different parasite profiles after treatment, including complete parasite suppression, parasite regrowth at different time points during follow-up, and initial parasite regrowth followed by later suppression, and could therefore describe most of the individual profiles of parasite recrudescence during follow-up (Fig A and Fig C in S1 File). High variability in recrudescence profiles was reflected in a very large and non-normally distributed BSV (>200 CV%) for baseline parasite load, I_{max} , and IT_{50} . Typical value plots, as depicted in Fig 5B, illustrate the effect of different onset times of immune suppression (different IT₅₀) on the parasite dynamics during follow-up. None of the haematological markers correlated with parasitological response (Fig D in S1 File) and white blood cell counts were not a significant covariate for k_{imm} or IT₅₀. Profiles with complete parasite clearance during treatment, followed by a fast and strong parasite recrudescence, could not be fully captured by the model (Fig A in S1 File), despite the restriction of low parasite loads to ≥ 1 parasites/mL in the model.

3.3 Correlation between parasite exposure and clinical outcome

The AUC_{ROC} for blood parasite load classifying clinical relapse (Fig 6) was highest on Day 28 (0.82) and Day 56 (0.87), compared with Day 10 (0.64) (Table 3). The optimal cutoff values for classifying relapse were around 10 parasites/mL (9 parasites/mL on Day 28 and 11 parasites/mL on Day 56). The fraction of patients having a blood parasite load <10 parasites/mL on Day 28 and Day 56 was therefore compared between treatment regimens and the final clinical outcome (Table 4). There were considerably more patients (73–86%) with an adequate parasite clearance (parasite load <10 parasites/mL) on Day 56 in relatively effective treatment regimens (AmB+SSG10D, AmB+MF10D, MFC28D, MFA28D), compared to only 10% in the non-efficacious Fexi10D regimen. Based on the cutoff value of 10 parasites/mL, 74% and 76% of relapsed patients were correctly categorized as relapse on Day 28 and Day 56, respectively.





Various other individual model-based secondary blood parasite load metrics were compared between treatment regimens and the final clinical outcome (Table 4). In the miltefosine monotherapy regimens (MFC28D and MFA28D), the individual predicted blood parasite loads were higher on Day 10 compared to other regimens, but reached and remained low blood parasite loads on Day 28 and Day 56 suggesting high overall efficacy. In the Fexi10D

Tuble 5. Terrormance of biolog parashe four on anterent augs areer reaction for ennear reacys, based on No 5 analysis,						
Day ^a	AUC _{ROC} ^b	Cutoff value (parasites/mL) ^c	Sensitivity	Specificity	Specificity	
10	0.64	51	0.44	0.85		
28	0.82	9	0.77	0.82		
56	0.87	11	0.88	0.77		

Table 3. Performance of blood parasite load on different days after treatment as a predictor for clinical relapse, based on ROC analysis.

^a Day after start of treatment

^b Area under the ROC curve

^c Optimal cutoff value of blood parasite load to predict clinical relapse

https://doi.org/10.1371/journal.pntd.0012078.t003

Study	LEAP0208			LEAP0714	FEXI-VL-001			
Treatment/outcome	AmB+SSG10D	AmB+MF10D	MFC28D	MFA28D	Fexi10D	Cure	Relapse	
Patients ^a (n)	40	44	46	29	13	138	34	
Cured patients (n [%])	37 (93)	37 (84)	35 (76)	27 (93)	2 (15)	138 (100)	0 (0)	
Parasite load <10p/mL at D28 (n [%])	31 (78)	31 (70)	32 (70)	18 (62)	3 (10)	106 (77)	9 (26)	
Parasite load <10p/mL at D56 (n [%])	33 (83)	32 (73)	34 (74)	25 (86)	3 (10)	119 (86)	8 (24)	
Parasite load D1 ^b (p/mL)	23802 (4904– 51372)	4493 (553– 34633)	4583 (1008– 29194)	33363 (6091– 77096)	3732 (2605– 14467)	7142 (1231– 40157)	14467 (4493– 62465)	
Parasite load D10 ^b (p/mL)	11.8 (2.8–35.9)	137 (4.0-807)	2027 (226– 12355))	2378 (429–44318)	105 (25.1–249)	150 (8.8–2094)	333 (122–5985)	
Parasite load D28 ^b (p/mL)	1.1 (1.0-6.1)	2.1 (1.0-22.9)	1.1 (1.0–135)	1.3 (1.1-42.2)	112 (18.2–216)	1.1 (1.0-6.1)	105 (11.6–313)	
Parasite load D56 ^b (p/mL)	1.0 (1.0-5.7)	1.1 (1.0–10.1)	1.3 (1.0-10.4)	1.2 (1.1-3.1)	486 (11.1-1073)	1.1 (1.0-2.7)	125 (11.9–613)	
Parasite AUC _{D0-10} ^b (p·day/ mL)	7284 (2385– 19935)	5970 (641– 41824)	32509 (8631– 162007)	39558 (7638– 417250)	8482 (4827– 36350)	15648 (3241– 52869)	17209 (5012– 97125)	
Parasite AUC _{D0-28} ^b (p·day/ mL)	7371 (2537– 20370)	7061 (804– 47564)	56327 (18322– 223201)	74829 (14150– 604792)	10260 (5644– 37750)	18642 (3382– 80241)	24596 (7042– 129150)	
Parasite AUC _{D0-56} ^b (p∙day/ mL)	7400 (3094– 20972)	10905 (906– 52583)	56906 (18973– 223236)	75175 (15254– 604833)	33768 (8532– 69033)	18884 (3464– 85774)	41931 (14171– 173361)	
I _{max} (day ⁻¹) ^c	2.3 (1.7)	1.9 (2.0)	1.4 (1.6)	1.2 (0.7)	0.9 (0.7)	1.9 (1.7)	0.7 (0.5)	
IT ₅₀ (day) ^c	23 (21)	45 (58)	59 (73)	60 (50)	77 (57)	40 (51)	84 (65)	

Table 4. Individual model-based predictions of parasite exposure per treatment arm and clinical outcome.

AmB+MF10D: Amphotericin B 10 mg/kg (1 day) + miltefosine 2.5 mg/kg (10 days); AmB+SSG10D: Amphotericin B 10 mg/kg (1 day) + SSG 20 mg/kg (10 days); BLQ: below the limit of quantification; Fexi10D: Fexinidazole 1800 mg (4 days), 1200 mg (6 days); I_{max}: maximum inhibitory effect of the immune system; IT₅₀: time after treatment of half the maximal immune effect; MFA28D: miltefosine allometric dosing (pediatrics); MFC28D: Miltefosine conventional dosing (2.5 mg/kg) (adults). p/mL: parasites/mL ^a Patients included in the analysis

^b Median (IQR)

^c Mean (sd). I_{max} and IT₅₀ represent the activity of the immune system in suppressing parasites

https://doi.org/10.1371/journal.pntd.0012078.t004

treatment arm, individual predicted blood parasite loads were higher on Day 28 and Day 56, which had poor efficacy. Blood parasite loads on Day 10, Day 28, and Day 56 were substantially higher in relapsed patients (Table 4 and Fig 7).

Although a low parasite AUC for different durations (AUC_{D0-10}, AUC_{D0-28}, AUC_{D0-56}) was not associated with favorable clinical outcome (also not when stratified by treatment regimen), a slightly lower AUC_{D0-56} of 18884 (3464–85774) p·day/mL was observed in cured patients compared to 41931 (14171–173361) p·day/mL in relapsed patients. The parasite AUCs were considerably higher in the miltefosine monotherapy regimens (MFC28D and MFA28D), which is in accordance with the slow miltefosine drug accumulation and slow onset of miltefosine-induced parasite clearance. The extent of suppression of parasite regrowth after treatment (I_{max}) did not clearly correlate with treatment efficacy, but the onset of the suppressive immune response (IT_{50}) was substantially delayed in the Fexi10D treatment arm, which had poor efficacy, indicating that a weak drug effect and partial parasite clearance during treatment negatively influenced the onset of the immune response. This was in accordance with the delayed IT₅₀ in relapsed patients compared to cured patients.

4. Discussion

A semi-mechanistic pharmacokinetic-pharmacodynamic model was developed, based on pharmacokinetic and blood parasite load data from Eastern African VL patients receiving five





different (combination or monotherapy) treatment regimens, to characterize the complicated interaction between parasite replication, drug-induced parasite clearance, and parasite clearance due to an emerging host immune response. The model adequately captured the blood parasite dynamics during and after treatment and revealed that blood parasite loads higher than 10 parasites/mL on Day 28 and Day 56 after start of treatment are an early indication of VL relapse, which could serve as a biomarker to predict long-term clinical outcome based on the sensitivity and specificity. Until now, it has not been possible to predict relapse, a long-term event that can occur up to 12 months or even longer after treatment. Moreover, the model indicated that the long-term clinical outcome depends both on the initial parasite clearance by the treatment and on parasite suppression after treatment, which is probably achieved by the host immunological response. A better understanding of this immunological response and its associated host biomarkers could potentially lead to an improved prediction of relapse.

Parasite replication and drug-induced parasite clearance by five different VL drug regimens were adequately characterized by the pharmacokinetic-pharmacodynamic model we developed. To our knowledge, the model provided the first-ever estimation of the *in vivo Leish-mania* parasite doubling time in human of 7.8 days, which was only slightly slower than previously reported *in vitro* intracellular *L. donovani* amastigote replication rates within macrophages, which correspond to parasite doubling times of 4.2 and 2.9 days [35,36]. However, one limitation of this analysis is the lack of data on 'natural' parasite growth prior to initiation of treatment. Parasite growth was estimated based on patient data after treatment (Fexi10D), where the estimated parasite growth might have been affected by the treatment itself or selection of certain parasite subpopulations due to the treatment. Moreover, identification of parasite growth is also dependent on the sampling scheme, and parasite growth rate might be variable among patients, depending on host-related factors such as the activity of the immune system.

The unusual PK characteristics of liposomal amphotericin B, complicate the construction of a PK/PD model for this particular drug, since it is unknown to what extent systemic liposomal amphotericin B concentrations (or dose kinetics) are actually driving parasite clearance. Liposomal amphotericin B is preferentially taken up by blood monocytes and tissue macrophages, in which *Leishmania* parasites primarily reside. This assumption, in combination with the absence of monotherapy data for both liposomal amphotericin B and SSG, may have contributed to the minor underprediction of parasite clearance observed in the AmB-based regimens.

The combination regimens liposomal amphotericin B with SSG or miltefosine led to rapid drug-induced parasite clearance, resulting in undetectable blood parasite loads by the end of drug exposure in most patients (70–78% of patients had a parasite load <10 parasites/mL on Day 28). Miltefosine monotherapy resulted in a slower or delayed onset of parasite clearance, in line with the slow accumulation of miltefosine in the first week of treatment. Miltefosine treatment outcome has been shown to be associated with the time above a target concentration [37]. On the other hand, the long half-life of the drug results in adequate above-target exposure, which extends for a considerable period of time after the end of the 28-day monotherapy (MFC28D and MFA28D) and results in complete parasite clearance by Day 56 (parasite load <10 parasites/mL on Day 56 in 74–86% of patients). Fexinidazole treatment resulted in only partial drug-induced parasite clearance in most patients, with only 2/13 patients having complete parasite clearance by the end of treatment, as measured by qPCR. The results of this model suggest that adequate parasite clearance by the drug is important for achieving longterm clinical cure, as almost all patients on fexinidazole therapy who had incomplete drug clearance (by the treatment) failed therapy. This was confirmed by the ROC analysis, as the optimal blood parasite load cutoff values for classifying clinical relapse were around 10 parasites/mL, meaning that very low parasite levels by the end of treatment are already associated with higher risk of clinical relapse. This was in line with previous results of a descriptive analysis of these data, concluding that the blood parasite load on Day 56 was the best time to predict clinical outcome, with a comparable optimal cutoff value of 20 parasites/mL [11].

Parasite suppression after treatment, presumably driven by the host immune system, was empirically described in the model by an E_{max} function with variable onset and magnitude of parasite clearance. Parasite dynamics after treatment were highly variable among patients. The model suggested that patients with complete parasite suppression had a fast onset of the host immune response, occurring relatively rapidly after end of treatment, while patients with parasite recrudescence somewhere during follow-up had a late onset or weak magnitude of the immune response. Although we used a sophisticated analysis method to incorporate the complex interplay between parasites, drugs, and host, the haematological and biochemical data explored could not explain the between-patient variability in onset and magnitude of posttreatment suppression or, conversely, recrudescence of parasites.

Because of the multitude of effects that play a role in parasite dynamics, such as parasite growth, simultaneous parasite clearance by the treatment, and parasite suppression by the immune system, simultaneous estimation of all parameters led to over-parameterization of the model. However, by estimating the parasite growth rate, the drug effects, and the host immune effect separately based on representative subsets of the data, we could adequately estimate a parasite growth rate and drug effects for all drugs. Precise parameter estimates of the immune response model were, nevertheless, difficult to obtain. To improve the stability and the convergence of the model, various estimation methods were attempted, as well as implementing a mixture model with different k_{imm} 's for cured and relapsed patients, but this did not result in stable parameter estimates. In future studies, more frequent blood sampling, especially during the follow-up phase, in combination with the identification of relevant host biomarkers, might improve the characterization of host suppression of the parasite load after treatment. While reinfection cannot be formally excluded, relapse is more likely as a cause of VL recurrence within the time-frame of the conducted clinical trials [38].

To predict parasite response after treatment, attempts were made to identify biomarkers for the activity of the immune system, by explaining variability in k_{imm}, but none could be identified. Identification of a biomarker for activity of the immune system is complicated because there are many factors involved. Cytokines and other direct immunological biomarker measurements were not available from the clinical trials currently included. White blood cell count was evaluated as a predictor for the activity of the host immune system, but was not relevant. In cured patients, lymphocyte and albumin levels were higher and total protein levels were lower, although these results should be interpreted with caution as they were only available for a small proportion of patients (Fig C in S1 File). Depletion of lymphocytes and albumin is a symptom of VL infection, and could therefore reflect severity of disease, and subsequently the ability of the patient to clear the parasite. Moreover, lymphocyte count might directly reflect the function of the immune system, as a Th1 response is probably responsible for activation of infected macrophages that can lead to intracellular L. donovani killing [39], once there are enough active T cells [10]. It would be worthwhile to collect blood cell counts and albumin levels in patients during treatment and follow-up in future clinical trials, as well as exploring other biomarkers related to the immune response, to further investigate this relationship using the developed PK/PD modeling framework.

Individual model-based predictions of parasite AUC and parasite loads at different time points were compared to the final clinical outcome. Parasite load at start of treatment was not related to relapse, indicating that the severity of infection at start of treatment is not indicative for the final treatment outcome. Parasite AUC was not predictive for clinical outcome either, which might be due to the aforementioned differences in pharmacokinetics between the treatment regimens. For example, the onset of parasite clearance due to miltefosine is much slower due to the initial slow accumulation of miltefosine, resulting in higher AUCs until Day 10, 28, and 56 for the miltefosine monotherapies compared to combination therapies with liposomal amphotericin B, but this treatment nevertheless resulted in a comparable final outcome. The parasite load on Day 28 and Day 56 after start of treatment was related to relapse, indicating that both an adequate drug response and an early onset of immune system activation are needed. Although the blood parasite load on Day 28 or Day 56 could not correctly predict relapse in all individual patients (26% and 24% of relapsed patients were classified as cured based on the ROC results, respectively), the fraction of patients having a blood parasite load <10 parasites/mL by the end of treatment in a group of patients receiving the same VL treatment is an indication of the efficacy of the treatment in this population. This might be useful

in clinical trials, where new dosing regimens or new combinations will be tested, to get an early indication of the long-term efficacy of this treatment, which would benefit from increased molecular biology capacity building in VL endemic areas in Eastern Africa. In the future, optimization of qPCR as a parasitological marker, possibly combined with other host biomarkers, could improve the sensitivity/specificity of a predictor for VL relapse, and be used as an endpoint in future VL trials to reduce the need for long periods of follow-up.

This is the first time that the complex mechanisms of parasite replication, treatment effects, and host response have been integrated into a comprehensive semi-mechanistic model, which provides insight into the *in vivo* parasite growth rate and parasite clearance rates by different drugs. The model could serve as a framework for optimization of VL treatment regimens, and blood parasite load at the end of treatment could be a useful biomarker to assess efficacy of treatment regimens in clinical trial settings.

Supporting information

S1 File. Supplementary data. (DOCX)

Acknowledgments

The authors thank all members of the field teams at the study sites, including clinicians, nurses and laboratory technicians, clinical monitors, and the Data Safety Monitoring Board, for their contributions to the studies. They warmly thank all the patients and the parents of the pediatric patients for their cooperation and willingness to be enrolled in this study. They also thank the local communities and acknowledge the Ministries of Health for their support.

Author Contributions

Conceptualization: Thomas P. C. Dorlo.

Data curation: Séverine Monnerat, Henk D. F. H. Schallig, Fabiana Alves.

Formal analysis: Luka Verrest, Wan-Yu Chu, Alwin D. R. Huitema, Fabiana Alves, Thomas P. C. Dorlo.

Project administration: Fabiana Alves.

Resources: Séverine Monnerat, Ahmed M. Musa, Jane Mbui, Eltahir A. G. Khalil, Joseph Olobo, Monique Wasunna, Henk D. F. H. Schallig.

Supervision: Thomas P. C. Dorlo.

Validation: Luka Verrest.

- Writing original draft: Luka Verrest, Alwin D. R. Huitema, Fabiana Alves, Thomas P. C. Dorlo.
- Writing review & editing: Séverine Monnerat, Ahmed M. Musa, Jane Mbui, Eltahir A. G. Khalil, Joseph Olobo, Monique Wasunna, Wan-Yu Chu, Henk D. F. H. Schallig, Thomas P. C. Dorlo.

References

1. Wasunna M, Njenga S, Balasegaram M, Alexander N, Omollo R, Edwards T, et al. Efficacy and Safety of AmBisome in Combination with Sodium Stibogluconate or Miltefosine and Miltefosine Monotherapy

for African Visceral Leishmaniasis: Phase II Randomized Trial. PLoS Negl Trop Dis. 2016;1–18. https://doi.org/10.1371/journal.pntd.0004880 PMID: 27627654

- Mbui J, Olobo J, Omollo R, Solomos A, Kip AE, Kirigi G, et al. Pharmacokinetics, Safety, and Efficacy of an Allometric Miltefosine Regimen for the Treatment of Visceral Leishmaniasis in Eastern African Children: An Open-label, Phase II Clinical Trial. Clinical Infectious Diseases. 2019; 68(9).
- Musa AM, Mbui J, Mohammed R, Olobo J, Ritmeijer K, Alcoba G, et al. Paromomycin and Miltefosine Combination as an Alternative to Treat Patients with Visceral Leishmaniasis in Eastern Africa: A Randomized, Controlled, Multicountry Trial. Clinical Infectious Diseases. 2022 Sep 27;
- Rijal S, Ostyn B, Uranw S, Rai K, Bhattarai NR, Dorlo TPC, et al. Increasing failure of miltefosine in the treatment of kala-azar in nepal and the potential role of parasite drug resistance, reinfection, or noncompliance. Clinical Infectious Diseases. 2013; 56(11):1530–8. https://doi.org/10.1093/cid/cit102 PMID: 23425958
- Gorski S, Collin SM, Ritmeijer K, Keus K, Gatluak F, Mueller M, et al. Visceral leishmaniasis relapse in Southern Sudan (1999–2007): A retrospective study of risk factors and trends. PLoS Negl Trop Dis. 2010 Jun; 4(6). https://doi.org/10.1371/journal.pntd.0000705 PMID: 20544032
- Burza S, Croft SL, Boelaert M. Leishmaniasis. Vol. 392, The Lancet. Lancet Publishing Group; 2018. p. 951–70.
- Murray HW, Hariprashad J, Fichtl RE, Murray HW. Models of Relapse of Experimental Visceral Leishmaniasis. The Journal ofInfectious Diseases. 1996; 173:1041–4. <u>https://doi.org/10.1093/infdis/173.4.</u> 1041 PMID: 8603949
- 8. Leta S, Dao THT, Mesele F, Alemayehu G. Visceral Leishmaniasis in Ethiopia: An Evolving Disease. PLoS One. 2014 Mar 1; 8(9). https://doi.org/10.1371/journal.pntd.0003131 PMID: 25188253
- Varma N, Naseem S. Hematologic changes in visceral Leishmaniasis/Kala Azar. Indian Journal of Hematology and Blood Transfusion. 2010; 26(3):78–82. <u>https://doi.org/10.1007/s12288-010-0027-1</u> PMID: 21886387
- Rodrigues V, Cordeiro-Da-Silva A, Laforge M, Silvestre R, Estaquier J. Regulation of immunity during visceral Leishmania infection. Parasit Vectors. 2016; 9(1):1–13. <u>https://doi.org/10.1186/s13071-016-1412-x PMID: 26932389</u>
- Verrest L, Kip AE, Musa AM, Schoone GJ, Schallig HDFH, Mbui J, et al. Blood Parasite Load as an Early Marker to Predict Treatment Response in Visceral Leishmaniasis in Eastern Africa. Clinical Infectious Diseases. 2021; 73(5):775–82. https://doi.org/10.1093/cid/ciab124 PMID: 33580234
- Bossolasco S, Gaiera G, Olchini D, Gulletta M, Martello L, Bestetti A, et al. Erratum: Real-Time PCR Assay for Clinical Management of Human Immunodeficiency Virus-Infected Patients with Visceral Leishmaniasis. J Clin Microbiol. 2004; 42(4):1858.
- Cascio A, Calattini S, Colomba C, Scalamogna C, Galazzi M, Pizzuto M, et al. Polymerase chain reaction in the diagnosis and prognosis of Mediterranean visceral leishmaniasis in immunocompetent children. Pediatrics. 2002;109(2).
- Cruz I, Cañavate C, Rubio JM, Morales MA, Chicharro C, Laguna F, et al. A nested polymerase chain reaction (Ln-PCR) for diagnosing and monitoring Leishmania infantum infection in patients co-infected with human immunodeficiency virus. Trans R Soc Trop Med Hyg. 2002; 96:S185–9. https://doi.org/10. 1016/s0035-9203(02)90074-x PMID: 12055836
- Disch J, Oliveira MC, Orsini M, Rabello A. Rapid clearance of circulating Leishmania kinetoplast DNA after treatment of visceral leishmaniasis. Acta Trop. 2004; 92(3):279–83. <u>https://doi.org/10.1016/j.</u> actatropica.2004.08.002 PMID: 15533299
- Fisa R, Riera C, Ribera E, Gállego M, Portús M. A nested polymerase chain reaction for diagnosis and follow-up of human visceral leishmaniasis patients using blood samples. Trans R Soc Trop Med Hyg. 2002; 96:S191–4. https://doi.org/10.1016/s0035-9203(02)90075-1 PMID: 12055837
- Kip AE, Balasegaram M, Beijnen JH, Schellens JHM, Vries PJ De, Dorlo TPC. Systematic Review of Biomarkers To Monitor Therapeutic Response in. Antimicrob Agents Chemother. 2015; 59(1):1–14.
- Lachaud L, Dereure J, Chabbert E, Reynes J, Mauboussin JM, Oziol E, et al. Optimized PCR using patient blood samples for diagnosis and follow-up of visceral leishmaniasis, with special reference to AIDS patients. J Clin Microbiol. 2000; 38(1):236–40. https://doi.org/10.1128/JCM.38.1.236-240.2000 PMID: 10618093
- Mary C, Faraut F, Drogoul MP, Xeridat B, Schleinitz N, Cuisenier B, et al. Reference values for Leishmania infantum parasitemia in different clinical presentations: Quantitative polymerase chain reaction for therapeutic monitoring and patient follow-up. American Journal of Tropical Medicine and Hygiene. 2006; 75(5):858–63. PMID: 17123977

- Mary C, Faraut F, Lascombe L, Dumon H. Quantification of Leishmania infantum DNA by a real-time PCR assay with high sensitivity. J Clin Microbiol. 2004; 42(11):5249–55. https://doi.org/10.1128/JCM. 42.11.5249-5255.2004 PMID: 15528722
- Nuzum E, White F, Thakur C, Dietze R, Wages J, Grogl M, et al. Diagnosis of symptomatic visceral leishmaniasis by use of the polymerase chain reaction on patient blood. Journal of Infectious Diseases. 1995; 171(3):751–4. https://doi.org/10.1093/infdis/171.3.751 PMID: 7876635
- Pizzuto M, Piazza M, Senese D, Scalamogna C, Calattini S, Corsico L, et al. Role of PCR in Diagnosis and Prognosis of Visceral Leishmaniasis in Patients Coinfected with Human Immunodeficiency Virus Type 1. J Clin Microbiol. 2001; 39(1):357–61. https://doi.org/10.1128/JCM.39.1.357-361.2001 PMID: 11136800
- Sudarshan M, Weirather JL, Wilson ME, Sundar S. Study of parasite kinetics with antileishmanial drugs using real-time quantitative PCR in Indian visceral leishmaniasis. Journal of Antimicrobial Chemotherapy. 2011; 66(8):1751–5. https://doi.org/10.1093/jac/dkr185 PMID: 21609983
- Cota GF, de Sousa MR, de Assis TSM, Pinto BF, Rabello A. Exploring prognosis in chronic relapsing visceral leishmaniasis among HIV-infected patients: Circulating Leishmania DNA. Acta Trop. 2017; 172 (December 2016):186–91. https://doi.org/10.1016/j.actatropica.2017.05.011 PMID: 28501450
- Abd-Rahman AN, Marquart L, Gobeau N, Kümmel A, Simpson JA, Chalon S, et al. Population Pharmacokinetics and Pharmacodynamics of Chloroquine in a Plasmodium vivax Volunteer Infection Study. Clin Pharmacol Ther. 2020 Nov 1; 108(5):1055–66. https://doi.org/10.1002/cpt.1893 PMID: 32415986
- Krause A, Dingemanse J, Mathis A, Marquart L, Möhrle JJ, McCarthy JS. Pharmacokinetic/pharmacodynamic modelling of the antimalarial effect of Actelion-451840 in an induced blood stage malaria study in healthy subjects. Br J Clin Pharmacol. 2016;412–21.
- 27. Wattanakul T, Baker M, Mohrle J, Mcwhinney B, Hoglund RM, Mccarthy JS, et al. Semimechanistic Pharmacokinetic and Pharmacodynamic Modeling of Piperaquine in a Volunteer Infection Study with Plasmodium falciparum Blood-Stage Malaria. Antimicrob Agents Chemother. 2021; 65(4):e01583–20. https://doi.org/10.1128/AAC.01583-20 PMID: 33468477
- Tarning J, Thana P, Phyo AP, Lwin KM, Hanpithakpong W, Ashley EA, et al. Population pharmacokinetics and antimalarial pharmacodynamics of piperaquine in patients with Plasmodium vivax Malaria in Thailand. CPT Pharmacometrics Syst Pharmacol. 2014; 3(8). <u>https://doi.org/10.1038/psp.2014.29</u> PMID: 25163024
- Palic S, Kip AE, Beijnen JH, Mbui J, Musa A, Solomos A, et al. Characterizing the non-linear pharmacokinetics of miltefosine in paediatric visceral leishmaniasis patients from Eastern Africa. J Antimicrob Chemother. 2020; 75:3260–8. https://doi.org/10.1093/jac/dkaa314 PMID: 32780098
- Dorlo TPC, Hillebrand MJX, Rosing H, Eggelte TA, de Vries PJ, Beijnen JH. Development and validation of a quantitative assay for the measurement of miltefosine in human plasma by liquid chromatography-tandem mass spectrometry. J Chromatogr B Analyt Technol Biomed Life Sci. 2008 Apr; 865(1– 2):55–62. https://doi.org/10.1016/j.jchromb.2008.02.005 PMID: 18325856
- Tarral A., Blesson S., Mordt O.V. et al. Determination of an Optimal Dosing Regimen for Fexinidazole, a Novel Oral Drug for the Treatment of Human African Trypanosomiasis: First-in-Human Studies. Clin Pharmacokinet 53, 565–580 (2014). https://doi.org/10.1007/s40262-014-0136-3 PMID: 24535888
- 32. Bekersky I, Fielding RM, Dressler DE, Lee JW, Buell DN, Walsh TJ. Plasma protein binding of amphotericin B and pharmacokinetics of bound versus unbound amphotericin B after administration of intravenous liposomal amphotericin B (AmBisome) and amphotericin B deoxycholate. Antimicrob Agents Chemother. 2002; 46(3):834–40. https://doi.org/10.1128/AAC.46.3.834-840.2002 PMID: 11850269
- Chulay JD, Fleckenstein L, Smith DH. Pharmacokinetics of antimony during treatment of visceral leishmaniasis with sodium stibogluconate or meglumine antimoniate. Trans R Soc Trop Med Hyg. 1988; 82 (1):69–72. PMID: 2845611
- Cruz A, Rainey PM, Herwaldt BL, Stagni G, Palacios R, Trujillo R, et al. Pharmacokinetics of antimony in children treated for leishmaniasis with meglumine antimoniate. Journal of Infectious Diseases. 2007; 195(4):602–8. https://doi.org/10.1086/510860 PMID: 17230422
- Tegazzini D, Díaz R, Aguilar F, Peña I, Presa JL, Yardley V, et al. A Replicative In Vitro Assay for Drug Discovery against Leishmania donovani. Antimicrob Agents Chemother. 2016; 60(6):3524–32. <u>https:// doi.org/10.1128/AAC.01781-15 PMID: 27021313</u>
- Castilla JJ, Sanchez-Moreno M, Mesa C, Osuna A. Leishmania donovani: In vitro culture and [1H] NMR characterization of amastigote-like forms. Mol Cell Biochem. 1995; 142(2):89–97. <u>https://doi.org/10. 1007/BF00928929 PMID: 7770070</u>
- Dorlo TPC, Kip AE, Younis BM, Ellis SJ, Alves F, Beijnen JH, et al. Visceral leishmaniasis relapse hazard is linked to reduced miltefosine exposure in patients from Eastern Africa: A population pharmacokinetic/pharmacodynamic study. Journal of Antimicrobial Chemotherapy. 2017; 72(11):3131–40. <u>https://</u> doi.org/10.1093/jac/dkx283 PMID: 28961737

- 38. Rijal S, Ostyn B, Uranw S, Rai K, Bhattarai NR, Dorlo TP, Beijnen JH, et al. Increasing failure of miltefosine in the treatment of Kala-azar in Nepal and the potential role of parasite drug resistance, reinfection, or noncompliance. Clin Infect Dis. 2013 Jun; 56(11):1530–8. https://doi.org/10.1093/cid/cit102 PMID: 23425958
- McElrath MJ, Murray HW, Cohn ZA. The dynamics of granuloma formation in experimental visceral leishmaniasis. Journal of Experimental Medicine. 1988; 167(6):1927–37. <u>https://doi.org/10.1084/jem.</u> 167.6.1927 PMID: 3290381